%\nocite{foley2021ad-software-engineering}
\nocite{sudermann2021mobile}
%\nocite{foley2021dindesign}
\nocite{des2021}
%\nocite{foley2021descomplexitykitchen}
%\nocite{foley2019complexitykitchen}
\nocite{icad2018book}
%\nocite{vossebeld2018customer}%01015
%\nocite{helgason2018tablet}%01015
\nocite{foley2017antenna}%01015
%\nocite{foley2017faucet}%01009
%\nocite{foley2017motorcyclelift}%01008
%\nocite{foley2017adjustadesk}%01002
\nocite{breznik2017collaborative}
\nocite{foley2017msddontology}
\nocite{petursson2017smaspring}
%\nocite{erlingsson2016roboclaw}
\nocite{foley2016desirable}
\nocite{guls2016auvlight}
%\nocite{gerhard2016suretrack}
\nocite{omarsdottir2016chessmate}
%\nocite{puik2016ignorance}
%\nocite{foley2016evaluating}
%\nocite{foley2016creative}
%\nocite{cochran2016hospital}
\nocite{cochran2016msdd}
\nocite{bragason2015parachuterelease}
\nocite{jonsson2015uget}
%\nocite{solvason2015spectrometer}
%\nocite{thompson2014additive}
\nocite{gudmundsdottir2014extendingviametricity}
%\nocite{gudmundsdottir2014wirelessscheduling}
%\nocite{foley2011colliding}
%\nocite{foley2012vitro}
\nocite{foley2012epd}
%\nocite{foley2012epdposter}
\nocite{kim2009microartificialmuscle}
\nocite{foley2008turbsim}
\nocite{engels2001networkedworld}

%%% Local Variables:
%%% mode: latex
%%% TeX-master: "foley-europass"
%%% End:
